Author's response to reviews

Title: Exuberant accessory mitral valve tissue with possible parachute-like asymmetric mitral valve in an asymptomatic adult patient: A case report

Authors:

Aleksandra Nikolić (nikolicdrsasa@gmail.com)
Zoran Joksimović (nikolicdrsasa@gmail.com)
Ljiljana Jovović (nikolicdrsasa@gmail.com)
Dejan Nikolic (nikolicdrsasa@gmail.com)

Version: 6 Date: 4 January 2012

Author's response to reviews:

Respected,

Comments to authors - reviewer 1

1. Title suggestion:

Reviewer 1. believes that the title should be: Exuberant accessory mitral valve tissue with possible parachute-like asymmetric mitral valve in an asymptomatic adult patient: A case report. We agree and have already changed the title. We suggest that you say “true parachute mitral valve” instead of “parachute-like”.

2. Quality of written English:

The text was revised by the official English lecturer. All changes were marked with “tracked changes” and uploaded.

3. Comment about key words:

We did not know where we should list all the keywords so we are listing them here: congenital heart disease, parachute mitral valvae, congenital mitral regurgitation.

Comments to authors - reviewer 2

1. Transoesophageal echocardiography was not done due to fact that patient refused. She has been abroad for a long while and is not available for an additional review. The patient refuses the fact that there is some medical problem and that she needs to have regular check-ups. The last information is that she will be back in the country in late January 2012, but we are not sure.
2. We do not have the MRI in our institution and the only institution that can give us information relevant to the publication is the University Children Hospital in Belgrade, which deals with the analysis of congenital heart diseases and which often cooperates with us in cases of CHD in adults. However, they have been operating (MRI) with long interruptions last year (this can be checked on their website) so we cannot get help from them at the moment. Unfortunately, these are the constraints that we have in Serbia when it comes to the diagnostics.

Grateful in advance

Authors